Author's response to reviews

Title: Quality of Health News Disseminated In the Print Media in Developing Countries; a Case Study in Iran

Authors:

Mahnaz Ashorkhani (ashoorkhani@farabi.tums.ac.ir)
Jaleh Gholami (jalehsahar@yahoo.com)
Katayoun Maleki (k_maleki@farabi.tums.ac.ir)
Sima Nedjat (nejat_s@farabi.tums.ac.ir)
Jalaledin Mortazavi (j_mortazavi@yahoo.com)
Reza Majdzadeh (rezamajd@tums.ac.ir)

Version: 5 Date: 20 July 2012

Author's response to reviews:

Dear Editor of the BMC Public Health,

The new revision is in response to your inquiry regarding conflict of interest of the authors and their contribution. I hope you find this revision suitable for publication.

Best regards,

RM